Author's response to reviews

Title: A qualitative analysis of the barriers and facilitators to HIV counselling and testing amongst adolescents in South Africa

Authors:

Michael R Strauss (watsupmike@gmail.com)
Bruce D Rhodes (rhodesb@ukzn.ac.za)
Gavin George (georgeg@ukzn.ac.za)

Version: 2 Date: 22 October 2014

Author's response to reviews:

Dear Editors

Please accept an updated version of our manuscript which better aligns with the journal's style. In this updated version, the main body of the manuscript remains unchanged, but a cover page has been included, along with sections for abbreviations, acknowledgements, competing interests, and author contributions. Line numbers have also been inserted as per formatting requirements.

Best,
Michael Strauss